Author’s response to reviews

Title: Validity and Reliability of the Greek Version of Implementation Leadership Scale (ILS)

Authors:

Eleni Mandrou (elenieleni101@gmail.com)
Andreas Tsounis (tsounis_a@yahoo.gr)
Pavlos Sarafis (pavlos.sarafis@cut.ac.cy)

Version: 2 Date: 18 Apr 2020

Author’s response to reviews:

TITLE: Validity and Reliability of the Greek Version of Implementation Leadership Scale (ILS)
PSYO-D-19-00259.

RESPONSE TO THE EDITOR

In the following, there are the comments and our responses which are marked with "RE".

Editor Comment: "In your Ethics approval section, please clarify whether consent obtained from participants was written or verbal. If consent was obtained verbally please include reasoning for this and state if an IRB approved this method of obtaining consent"

RE: Participants provided written informed consent. We added a relevant statement at the end of Ethics approval section.

Editor Comment: "Please elaborate on where the Availability of data and materials can be found"

RE: In the manuscript, we added a statement regarding availability of data and materials after the conclusion and before abbreviations.

Editor Comment: "At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files"

RE: Manuscript was uploaded according to your requirements (clean version with no tracked changes, comments, highlights, strikethroughs or text in different colors). Figures were uploaded as a separate file.